Author's response to reviews

Title: Effectiveness of DNA-recombinant anti-hepatitis B vaccines in blood donors: a cohort study.

Authors:

Emil J Kupek (kupek@ccs.ufsc.br)
Denise ER Souza (danierig@terra.com.br)
Andrea Petry (deapetry@yahoo.com.br)

Version: 4 Date: 8 October 2007

Author's response to reviews: see over
RESPONSE TO REVIEWER FRANSISCO SOUTO (second revision:)

Minor essential revisions

Reviewer: I have noticed that the term “HBsAg” has appeared in the text in a wrong graphy “HbsAg”. Conventionally, the “B” of this term should be appeared as cap.”

Authors: The correction suggested above has been made.